Dear Prof. Tarnawski

Thank you very much for reviewing our manuscript: Endoscopic treatment of scarred polyps with a nonthermal device (EndoRotor): A review of the literature. (ref 90165)

Thank you also to the reviewers. We have included a more detailed description of the Pros and Cons of Endorotor for the treatment of scarred polyps, and the changes have been highlighted. Also, our manuscript has been reviewed by a professional editing service – American Journal Experts. The certificate for the editing work has been uploaded. We hope that this revised version of our manuscript can be acceptable for publication in the World Journal of Gastroenterology.

Sincerely,

Adolfo Parra-Blanco
(on behalf of all the authors)